Author's response to reviews

Title: Role of respondents' education as a mediator and moderator in the association between childhood socio-economic status and later health and wellbeing.

Authors:

Mashhood A Sheikh (mashhood.a.sheikh@uit.no)
Birgit Abelsen (birgit.abelsen@uit.no)
Jan A Olsen (jan.abel.olsen@uit.no)

Version: 5  Date: 21 October 2014

Author's response to reviews: see over
Dear Editor-in-Chief,

Please find enclosed our revised manuscript, “Role of respondents’ education as a mediator and moderator in the association between childhood socio-economic status and later health and wellbeing.” by Sheikh, M.A., Abelsen, B., Olsen, J.A..

We are grateful to the two reviewers for their detailed comments and suggestions for improvements of the manuscript. Below you will find our response to the reviewer and the Journal Editorial Office (referred with line numbers in the manuscript). The text from the reviewers/ Journal Editorial Office is in blue.

Reviewer 2:

Reviewer's report:
1. I am satisfied that the authors have addressed the previous concerns, within the limitations of the data available.

We understand that the reviewer find the paper acceptable in its form. We are very thankful for the comments and suggestions.

Minor Essential Revision
2. Line 444: “odds ratio’s” should read “odds ratio”.

This has now been corrected.

Journal Editorial Office:
Furthermore, please be advised that manuscripts must include a ?Competing interests? section. This should be placed after the Conclusions/Abbreviations. If there are none to declare, please include the statement ?The authors declare that they have no competing interests.? Please consider the following questions and include an appropriate declaration of competing interests in your manuscript:

We have now included a “Competing interests” section (line 577-578).

We look forward to hearing from you at your earliest convenience.

Mashhood Ahmed Sheikh
Birgit Abelsen
Jan Abel Olsen